Author's response to reviews

Title: Applied public health research - falling through the cracks?

Authors:

Rebecca K Simmons (rebecca.simmons@mrc-epid.cam.ac.uk)
David Ogilvie (david.ogilvie@mrc-epid.cam.ac.uk)
Simon J Griffin (simon.griffin@mrc-epid.cam.ac.uk)
Lincoln A Sargeant (ls348@cam.ac.uk)

Version: 2 Date: 15 May 2009

Author's response to reviews: see over
15th May, 2009

Dear Dr Norton,

**Re: Applied public health research – falling through the cracks?**

With regards to an email received from Dr Andrea Bucceri (Scientific Editor) on 14th May 2009, we have uploaded a revised manuscript. We have now marked the manuscript as a *Correspondence article* (Type 3), rather than a *Debate article*, and have updated the Abstract accordingly.

We look forward to hearing from you.

Yours faithfully,

Dr Rebecca K. Simmons  
**Career Development Fellow**